Author's response to reviews

Title: Psychosocial factors and distress: A comparison between ethnic Norwegians and ethnic Pakistanis in Oslo, Norway

Authors:

Hammad R Syed (hammad@nakmi.no)
Odd S Dalgard (o.s.dalgard@medisin.uio.no)
Ingvild Dalen (ingvild.dalen@medisin.uio.no)
Bjorgulf Claussen (bjorgulf.claussen@medisin.uio.no)
Akthar Hussain (akhtar.hussain@medisin.uio.no)
Randi Selmer (rans@fhi.no)
Nora Ahlberg (n.i.ahlberg@medisin.uio.no)

Version: 5 Date: 3 July 2006

Author's response to reviews:

Covering letter

July 3, 2006

To
Editor-in-Chief
BMC Public Health
BioMed Central Ltd
Middlesex House
34-42 Cleveland Street
London W1T 4LB, UK.

From
Hammad Raza Syed
Institute of General Practice and Community Medicine (IASAM)
Department of International Community Health
P.O. Box 1130 Blindern
N-0318 Oslo, Norway

Dear Editor

Thanks for nice support and guidance during the review and submission process.

I will mention the changes made by me in the latest version of my manuscript submitted to your journal.

1. A few corrections were made in the background section of the abstract. First three sentences are added, modified and corrected.
2. A sentence about the written consent has been added under the section of research design. This is the last sentence under this section.
3. Abstract is also added in the manuscript file.
4. The title page and author information is also a part of manuscript now.
5. In the submission system, title page only provides information about the title of the manuscript. The author's information has been removed from here.

I hope that I have responded according to the possible suggestions sent to me. Please contact me again if you have some other suggestions regarding my manuscript.

Have a nice time during the sunny days of summers.

Regards
Hammad Raza Syed